Notes of the RICH1 Phone meeting 13/04/2007
People Present:
Bristol: Nick(part), Richard

CERN: Bill, Dave, Fabio, Kenneth, Werner Richard P.
IC:        Richard P. 
Apologies for absence:
Clara, Neville
Discussed Agenda Topics:
1. Gas Enclosure
Tooling for removing GE side doors has been brought out to CERN. It was fitted successfully and both doors were removed. The GE sides are now protected with plastic sheets. The upper H-frame for the flat mirror installation was added onto the A-side, while the lower H- frame needs an additional component to rectify 2 holes that have been drilled in the wrong position. The lower flat mirror support rail was installed and the spherical mirror load support rails were test installed in the GE.  

2. Spherical Mirrors
All completed mechanics parts were brought to CERN. Some pieces are still missing e.g. upper rails not yet ready. All installation tooling is at CERN. The spherical mirror support mechanics is not yet ready for a trail installation. The bridging rail needs to have ~0.2mm removed (action Tim).  The installation tooling will be checked in the pit.
3. Flat Mirrors
The majority of the support mechanics is now at CERN. The remaining parts are being machined at IC. The adjuster mechanism was delivered yesterday. The two lower support plates, the installation boxes and the platform are in 156. The roller balls, shackles, nuts have been ordered by Richard and Fabio. The welds inside the frame are not ground off resulting in an interference with the rail bracket. Tim is the whole of next week at CERN. He will machine the various parts which need correcting; rail bracket etc. (action Tim). 

If holes need to be drilled in the floor for the alignment survey these must be drilled before the tent is cleaned. (action Fabio).
4. HPD Assembly
Dressing: Top half completed and tested. The final 6 end rungs are at CERN. There are enough cups for 14 columns, however 26 are missing and it is not known where they are.
Rail System:   Received authorization from purchasing to go ahead with the order and it is under production at NAB. Delivery ~May07. PPSPS: waiting for calculations from Rohan (action Rohan).
Photon Funnel:  Drawings sent out to company, the same one as for RICH2. Brian Otwell is taking the design of the lower rail system; which is to be rotated around the z-axis when box is taken out.
Cabling: There is a large quantity of cables cut to length with the connectors attached. The upper tray is not yet installed.  Following installation of the tray the scaffolding will remain in position for the upper cable installation.

5.  MDMS
Had meeting with Ray. It is clear that the photon funnel could not be expanded to fit Ray’s system. Ray’s system has to fit within the constraints because the photon funnel is out for production. Allowing the light source to be asymmetric, giving coverage of 6 out of 7 rows, the system could fit.  Next meeting is 17 April07.
6. Aerogel
Next week the trail mounting of the aerogel boxes will be carried out. There is still concern about the window; the 300 μm thick glass could break damaging the beampipe. 

7. Next meeting
Friday 27th April 2007 at 10:00(GMT) 11:00(CET). 
